Author's response to reviews

Title: Quality of medication use in primary care: mapping the problem, working to a solution - a systematic review of the literature.

Authors:

Sara Garfield (saragarfield@yahoo.co.uk)
Nick Barber (n.barber@pharmacy.ac.uk)
Paul Walley (paul.walley@wbc.ac.uk)
Alan Wilson (alan.willson@nliah.wales.nhs.uk)
Lina Eliasson (lina.eliasson@pharmacy.ac.uk)

Version: 4 Date: 28 August 2009

Author's response to reviews: see over
Dear Robin,

Thank you for your email of 18th August accepting the paper provided we complied with one of the referees remaining two points. We have responded to the referee’s comments below.

1. The second paragraph of the background paragraph (beginning ‘It is generally’) has been split, as requested.

2. We have added examples of systematic techniques which have been used to improve healthcare processes, as requested.

We are very pleased that you have accepted the paper for BMC Medicines.

Yours sincerely,

Professor Nick Barber, The School of Pharmacy, University of London, Visiting Professor in Patient Safety, Harvard Medical School

Dr Sara Garfield, Postdoctoral Research Fellow, The School of Pharmacy, University of London